# Rituximab compared to intravenous cyclophosphamide in adults with connective tissue disease-associated interstitial lung disease: the RECITAL RCT

Toby M Maher,<sup>1,2\*</sup> Veronica A Tudor,<sup>2</sup> Peter Saunders,<sup>3</sup> Fernando Zanghelini,<sup>4</sup> Carlota Grossi Sampedro,<sup>4</sup> Georgios Xydopoulos,<sup>4</sup> Michael Gibbons,<sup>5</sup> Sophie V Fletcher,<sup>6</sup> Christopher P Denton,<sup>7</sup> Maria Kokosi, Rachel K Hoyles,<sup>3</sup> Helen Parfrey,<sup>8</sup> Elisabetta A Renzoni,<sup>2,9</sup> Athol U Wells,<sup>2,9</sup> Deborah Ashby,<sup>10,11</sup> Richard J Fordham,<sup>4</sup> Matyas Szigeti<sup>12,13</sup> and Philip L Molyneaux<sup>2,9</sup>; on behalf of the RECITAL Investigators

<sup>1</sup>Division of Pulmonary, Critical Care and Sleep Medicine, Keck School of Medicine, University of Southern California, Los Angeles, USA

- <sup>2</sup>Royal Brompton and Harefield Hospitals, Guy's and St Thomas' NHS Foundation Trust, London, UK
- <sup>3</sup>Oxford Centre for Respiratory Medicine, Churchill Hospital, Oxford, UK
- <sup>4</sup>Norwich School of Medicine, University of East Anglia, Health Economics Consultancy, Norwich Research Park, Norwich, UK
- <sup>5</sup>Academic Department of Respiratory Medicine, Royal Devon and Exeter Foundation Trust, Exeter, UK
- <sup>6</sup>NIHR Southampton Biomedical Research Centre, University Hospital Southampton, Southampton, UK
- <sup>7</sup>Centre for Rheumatology, Division of Medicine, Royal Free Campus, University College London, London, UK
- <sup>8</sup>Interstitial Lung Disease Unit, Royal Papworth, Hospital NHFT Cambridge Biomedical Campus, Cambridge, UK
- <sup>9</sup>National Heart and Lung Institute, Imperial College London, London, UK
- <sup>10</sup>School of Public Health, Imperial College London, London, UK
- <sup>11</sup>Imperial NIHR Biomedical Research Centre, London, UK
- <sup>12</sup>Imperial Clinical Trial Unit, Imperial College London, London, UK
- <sup>13</sup>Physiological Controls Research Centre, Obuda University, Budapest, Hungary

\*Corresponding author toby.maher@med.usc.edu

### **Disclosure of interests**

**Full disclosure of interests:** Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/LYWQ8541.

**Primary conflicts of interest:** Toby M Maher declares institutional grants from Astra Zeneca and GlaxoSmithKline R&D, and personal fees from Astra Zeneca, Bayer, Blade Therapeutics, Boehringer Ingelheim, Bristol-Myers Squibb, Chiesi, Fibrogen, Galapagos, Galecto, GlaxoSmithKline R&D, IQVIA, Pliant, Roche Trevi Therapeutics and Veracyte. Peter Saunders declares personal fees from Trevi Therapeutics and Boehringer Ingelheim.

Sophie V Fletcher declares personal fees from Chiesi and Boehringer Ingelheim.

Helen Parfrey declares personal fees from Boehringer Ingelheim, Trevi Therapeutics and Pliant Therapeutics. Athol U Wells declares personal fees from Boehringer Ingelheim, Roche and Veracyte. Philip L Molyneaux declares an institutional grant from Astra Zeneca and personal fees from Astra Zeneca, Boehringer Ingelheim and Roche.

Published February 2024 DOI: 10.3310/LYWQ8541

## **Plain language summary**

Rituximab compared to intravenous cyclophosphamide in adults with connective tissue disease-associated interstitial lung disease: the RECITAL RCT

Efficacy and Mechanism Evaluation 2024; Vol. 11: No. 4 DOI: 10.3310/LYWQ8541

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain language summary**

nterstitial lung disease, a condition characterised by inflammation and scarring of the lungs, is the leading cause of death in systemic sclerosis (an autoimmune disease that typically causes thickening and scarring of the skin and which is associated with internal organ problems such as interstitial lung disease and kidney failure), and a major cause of morbidity (illness) in many other connective tissue diseases; a group of conditions that are caused by over activity of the immune system.

When interstitial lung disease associated with connective tissue disease gets worse over time, treatment such as intravenous cyclophosphamide is required to slow down lung scarring. Occasionally, standard immunosuppressive drugs fail to control lung inflammation and scaring and this can result in death.

Rituximab, a novel therapy, has been proven to be of benefit in suppressing inflammation associated with immune system over activity. Observational studies suggest that rituximab may be an effective treatment for pulmonary inflammation in connective tissue diseases.

The study was designed to determine how well rituximab works compared to cyclophosphamide in treating patients with severe connective tissue disease-associated interstitial lung disease. We recruited 101 participants from 11 hospitals throughout the UK who were randomly allocated to one of two groups. Those in the first group were given rituximab on day 1 of the study and then on day 14. They were then given a placebo every 4 weeks for the next 18 weeks. Those in the second group were given cyclophosphamide every 4 weeks from day 1 of the study to week 20. On day 14, they were given a placebo. Lung function for all participants was assessed at 24 weeks.

Our results suggest that rituximab improved lung function and quality of life but was not better than cyclophosphamide. Rituximab was associated with fewer unexpected medical events and a trend towards reduction in corticosteroid use and should be considered as a therapeutic alternative to cyclophosphamide.

### **Efficacy and Mechanism Evaluation**

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

A list of Journals Library editors can be found on the NIHR Journals Library website

Efficacy and Mechanism Evaluation (EME) was launched in 2014 and is indexed by Europe PMC, DOAJ, Ulrichsweb<sup>™</sup> (ProQuest LLC, Ann Arbor, MI, USA) and NCBI Bookshelf.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme.

#### Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Manuscripts are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

### **EME** programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in humans and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health and Care Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

#### This manuscript

The research reported in this issue of the journal was funded by the EME programme as project number 11/116/03. The contractual start date was in September 2013. The final report began editorial review in April 2022 and was accepted for publication in February 2023. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' manuscript and would like to thank the reviewers for their constructive comments on the final manuscript document. However, they do not accept liability for damages or losses arising from material published in this manuscript.

This manuscript presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the EME programme or the Department of Health and Social Care.

Copyright © 2024 Maher *et al.* This work was produced by Maher *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).